Author's response to reviews

Title: Morphometric Variability of Neuroimaging Features in Children with Agenesis of the Corpus Callosum

Authors:

Richard Mayeux (rpm2@columbia.edu)
Jason B Neal (jbn2113@columbia.edu)
Christopher G Filippi (cf2529@columbia.edu)

Version: 2
Date: 13 January 2015

Author's response to reviews:

The manuscript is a research article rather than a case report, it has been re-formatted according to the research article form.